Author's response to reviews

Title: Efficacy and Cost-Effectiveness of minimal guided and unguided Internet-based Mobile supported Stress-management in Employees with Occupational Stress. A three-armed Randomised Controlled Trial.

Authors:

David D Ebert (ebert@leuphana.de)
Dirk Lehr (Lehr@inkubator.leuphana.de)
Filip Smit (fsmit@trimbos.nl)
Anna-Carlotta Zarski (zarski@inkubator.leuphana.de)
Heleen Riper (h.riper@vu.nl)
Elena Heber (heber@inkubator.leuphana.de)
Pim Cuijpers (p.cuijpers@vu.nl)
Matthias Berking (berking@fau.de)

Version: 2
Date: 16 July 2014

Author's response to reviews:

As requested from the editorial office we made it more clear in the manuscript that ethical approval was already given by the ethical comittee of the university of marburg.

We were also ask to report, that the study protocol detailed reviewed as part of the grant review process from experts of the funding agency and to upload the manuscript without the additional files such as ethical approval, funding grant, etc.